RESEARCH

Open Access



Aluma Chovel Sella^{1,2}, Kendra R. Becker^{3,4}, Meghan Slattery¹, Kristine Hauser¹, Elisa Asanza¹, Casey Stern³, Megan Kuhnle¹, Nadia Micali^{5,6,9}, Kamryn T. Eddy^{3,4}, Madhusmita Misra^{1,2,7,8†}, Jennifer J. Thomas^{3,4†} and Elizabeth A. Lawson^{1,8*†}

Abstract

Background Avoidant/restrictive food intake disorder (ARFID) is a restrictive eating disorder commonly associated with medical complications of undernutrition and low weight. In adolescence, a critical time for bone accrual, the impact of ARFID on bone health is uncertain. We aimed to study bone health in low-weight females with ARFID, as well as the association between peptide YY (PYY), an anorexigenic hormone with a role in regulation of bone metabolism, and bone mineral density (BMD) in these individuals. We hypothesized that BMD would be lower in low-weight females with ARFID than healthy controls (HC), and that PYY levels would be negatively associated with BMD.

Methods We performed a cross-sectional study in 14 adolescent low-weight females with ARFID and 20 HC 10–23 years old. We assessed BMD (total body, total body less head and lumbar spine) using dual x-ray absorptiometry (DXA) and assessed fasting total PYY concentration in blood.

Results Total body BMD Z-scores were significantly lower in ARFID than in HC (-1.41 ± 0.28 vs. -0.50 ± 0.25 , p = 0.021). Mean PYY levels trended higher in ARFID vs. HC (98.18 \pm 13.55 pg/ml vs. 71.40 \pm 5.61 pg/ml, p = 0.055). In multivariate analysis within the ARFID group, PYY was negatively associated with lumbar BMD adjusted for age ($\beta = -0.481$, p = 0.032).

Conclusion Our findings suggest that female adolescents with low-weight ARFID may have lower BMD than healthy controls and that higher PYY levels may be associated with lower BMD at some, but not all, sites in ARFID. Further research with larger samples will be important to investigate whether high PYY drives bone loss in ARFID.

Plain English summary

Avoidant/restrictive food Intake disorder (ARFID) is a condition characterized by lack of interest in eating/food, sensory sensitivity and/or fear of aversive consequences of eating. It is associated with low weight and undernutrition, which

[†]Madhusmita Misra, Jennifer J. Thomas and Elizabeth A. Lawson have shared senior authorship.

*Correspondence: Elizabeth A. Lawson ealawson@mgh.harvard.edu Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/A.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

can lead to medical complications. Specifically, low weight in patients with ARFID raises concerns of impaired bone health. In this study, we compared bone mineral density (BMD), a measure of bone health, in 14 low-weight females with ARFID and 20 healthy females 10–23 years old. We also examined the association between BMD and peptide YY (PYY), a hormone that induces satiety and inhibits bone formation. A strong negative association between bone health and PYY was previously reported in females with anorexia nervosa. Thus, we hypothesized a similar association in low weight females with ARFID. We found that BMD may be lower in low-weight females with ARFID than in healthy females and that higher PYY levels are associated with lower BMD at some but not all sites. We concluded that bone health may be a concern in low-weight females with ARFID. This finding is important as low BMD raises concerns for increased fracture risk, which in turn could have a detrimental effect on guality of life.

Keywords Adolescence, ARFID, Avoidant/restrictive food intake disorder, Bone health, Bone mineral Density, BMD, DXA, Feeding and eating disorder, Low weight, PYY

Introduction

Avoidant/restrictive food intake disorder (ARFID) is characterized by a lack of interest in eating or food, sensory sensitivity, and/or a fear of aversive consequences of eating; as opposed to the body image disturbance and fear of weight gain that characterize anorexia nervosa (AN) [1, 2]. Similar to AN, ARFID is associated with medical complications of malnutrition [3, 4]. AN carries a significant risk for multifactorial bone loss due to undernutrition and low weight [5, 6]. In one study, 41% of adolescents girls were found to have bone mineral density (BMD) Z-scores of less than -1 at any one site and an additional 11% had BMD Z-scores of less than -2 [5].

Adolescents with ARFID are commonly underweight and malnourished, similar to patients with AN [1, 7]. This raises concerns for negative effects on bone health, particularly because adolescence is a critical time for bone accrual with more than 80% of peak bone mass accrued by 18 years of age [8]. Little is known about BMD in ARFID, with only two studies to date exploring this topic. One showed lower BMD in low-weight adult males with ARFID versus healthy controls (HC)[9] and the second reported no significant lumbar BMD Z-scores differences between ARFID and AN. [10] However, no study thus far has compared young low-weight females with ARFID to HC. There is a need to address this knowledge gap.

Peptide YY (PYY) is a gut derived anorexigenic hormone linked to bone loss in AN [11, 12]. PYY is secreted primarily by neuroendocrine L cells of the distal gut in response to food ingestion and promotes satiety by binding to the hypothalamic Y2 receptors of neuropeptide Y [13]. In rodents, selective deletion of the PYY receptor, Y1, in osteoblasts or of the Y2 receptor in the hypothalamus resulted in high bone mass indicating that PYY is a negative regulator of bone via activation of these receptors [13–15]. Specifically, deletion of the Y1 receptor in osteoblasts showed increased osteoblastic bone formation and mineralization rates via increased Runx2 and Osterix expression [16]. Further, previous studies have shown that females with AN have elevated PYY levels compared to normal weight individuals or those with obesity [11, 12] and demonstrated a strong inverse correlation between PYY and BMD [11, 12]. We recently reported that fasting and postprandial levels of PYY in low-weight females with ARFID did not differ from HC or females with AN of similar low weight. Such PYY levels would be considered to be maladaptively high for a satiety inducing hormone in low-weight individuals, as decreased anorexigenic signaling would typically be expected in an undernourished state [17].

Given these findings, we speculated that bone health of low-weight females with ARFID may be compromised by undernutrition. Additionally, inappropriately high levels of PYY in the setting of undernutrition may contribute to bone loss. In this first cross-sectional study of BMD in young low-weight females with ARFID compared to healthy normal-weight controls, we hypothesized that low-weight females with ARFID would have lower BMD, and that in the ARFID group, higher PYY levels would be associated with lower BMD.

Methods

Design

Participants were drawn from National Institutes of Health (NIH) funded studies evaluating the neurobiology of eating disorders (R01MH103402, Apr 2014–Mar 2020; R01MH108595, Mar 2016–Feb 2021; participants with ARFID and HC), bone health in AN (R01DK062249, July 2003–Feb 2011; HC), and endocrine function in young athletes (R01HD60827, Sep 2009–Jun 2016; HC).

For participants < 18 years old, written consent was signed by a parent/guardian and assent by the participant. Participants \geq 18 years signed written consent. All study procedures were approved by the Mass General Brigham Institutional Review Board. Participants were seen at the Massachusetts General Hospital (MGH) Translational and Clinical Research Center and at the

MGH/Harvard-Massachusetts Institute of Technology Division of Health Science and Technology Martinos Center for Biomedical Imaging.

Subjects

We studied 14 low-weight females with ARFID and 20 HC aged 10–23 years. The diagnosis of ARFID was confirmed by one of the following:

- 1. The Kiddie Schedule for Affective Disorder and Schizophrenia—Present and Lifetime (KSADS-PL) [18] (for studies R01MH108595 and R01MH103402). This is a semi-structured interview that generates DSM-5 Axis I diagnoses, including feeding and eating disorders, for youths. In both studies, we used the KSADS to rule out eating disorders other than ARFID, and to assess restrictive eating behaviors consistent with ARFID.
- 2. The Eating Disorder Assessment for DSM-5 (EDA-5) (in study R01MH108595). This is a semi-structured interview specifically developed to derive DSM-5 feeding and eating disorder diagnoses, including ARFID [19].
- 3. The Pica, ARFID, and Rumination Disorder Interview (PARDI) [20] (in study R01MH108595). This is a semi-structured interview that can be used to confer ARFID diagnoses and assess severity and related impairment.
- 4. The Eating Disorder Examination (EDE) Version 17.0 [21] (in study R01MH103402). This is a semi-structured clinical interview used to confer DSM-5 feeding and eating disorder diagnoses. In R01MH103402 we used the EDE to rule out eating disorders other than ARFID, and to assess restrictive eating behaviors consistent with ARFID.

By study definition, all subjects with low-weight ARFID had \leq 90% of expected body weight (EBW) determined by body mass index (BMI) divided by 50th percentile of BMI for age; or \leq 90% of EBW for height [17] and did not meet criteria for any other eating disorder. HC were required to have a BMI between the 10th to 90th percentiles, no pubertal delay (pubertal delay defined as menarche at > 16 years or thelarche at > 13 years) and regular menstrual periods (i.e. 9 or more menses over 12 months), if \geq 2 years post-menarche.

Exclusion criteria for all participants included hematocrit < 30%, active pregnancy or current breastfeeding, use of systemic hormones or oral contraceptives within 8 weeks of enrollment, history of psychosis, active suicidal ideation, active substance or alcohol use disorder, intellectual disability (IQ < 70), any significant illness that the investigator determined could interfere with the study and impact data collection or participant safety, contraindications to MRI or inability to tolerate being in the MRI machine for an hour (due to other analyses done in these parent studies).

Additional exclusion criteria for HC (depending on the study they were drawn from) included potassium < 3 mmol/l, glucose < 50 mg/dl (R01 DK062249); elevated FSH (R01 DK062249 and R01 HD60827); use of medications or concurrent diseases known to affect bone metabolism (R01 DK062249 and R01 HD60827); bone fracture within 6 months of the study (R01 DK062249); history of migraines, thromboembolism, smoking and a first-degree relative with breast cancer (R01 HD60827); migraines with aura (R01MH103402); vegetarianism, familial history of anorexia nervosa or other low weight eating disorders in first degree relatives (R01MH103402); gastrointestinal surgeries, a lifetime history of psychiatric disorder by KSADS-PL, any feeding or eating disorder as assessed via EDA-5 or by history (R01MH108595, R01MH103402).

Study procedures

Following informed consent, participants were screened to determine eligibility. The screening and baseline visits included a detailed medical history, physical examination including measurements of height, weight, and Tanner staging, followed by a blood sample to rule out anemia, urine β HCG to rule out pregnancy and an assessment of psychopathology by the KSAD-PL, PARDI, or EDE (R01MH103402, R01MH108595).

For eligible participants, we performed an assessment of bone health by dual-energy x-ray absorptiometry (DXA) at the baseline visit or at a main study visit, or at a separate visit for DXA. All participants obtained a blood sample for PYY following an overnight fast.

Biochemical and bone mineral density analyses

To assess fasting plasma PYY levels, we collected whole blood in EDTA-plasma tubes and placed on ice. Plasma was separated and stored at -80 °C. PYY levels were analyzed at the Brigham Assay Core Laboratory using an enzyme linked immunosorbent assay [Millipore Corporation; intra-assay coefficient of variation (CV) 17–18%, inter-assay CV 12–18%, lowest reportable value 10 pg/ml with dynamic range 10–2000 pg/ml)].

We assessed BMD and BMD Z-scores of the lumbar spine, total body less head, and total body, the three recommended sites for evaluating bone health in children and adolescents, by DXA using Hologic QDR-Horizon A, software version 13.6.0.4 and 13.6.0.5; Hologic Inc., Waltham, Massachusetts (MA), and by the Hologic QDR-Discovery A, software versions 13.3 and 13.5.3.2; Hologic Inc., Waltham, MA. From DXA reports, we extracted BMD Z-scores for the lumbar spine and total body based on means and standard deviations for age, sex and race available in the Hologic and Discovery A database. This information was not available for the total body less head for all subjects. Thus, BMD Z-scores for total body less head were obtained using the Zemel calculator based on data from the longitudinal Bone Mineral Density in Childhood Study [22]. Because this calculator allows BMD Z-score assessment for individuals \leq 20 years old only, total body less head BMD Z-scores are not available for 3 ARFID and 4 HC participants, who were \geq 21 years old.

Statistical analysis

We compared demographic and clinical characteristics across the ARFID and HC groups using the Student *t*-test (as all variables were normally distributed) for continuous variables and Chi Square for categorial variables. We present all continuous variables as mean \pm standard error of the mean (SEM) and all categorical data as count (%). We used linear regression to determine associations between BMD and PYY levels in low weight females with ARFID, followed by multivariate analysis to control for age. We defined statistical significance as a two-tailed *p*-value <0.05. We performed statistical analyses using JMP Pro 16.0.0 software.

Results

Table 1 displays demographics and clinical characteristics. Low-weight females with ARFID and HC were 16.41 ± 0.96 and 16.29 ± 0.78 years old, respectively (p = 0.954), with no significant difference in Tanner staging for breasts or pubic hair $(3.71 \pm 0.30 \text{ vs} 4.05 \pm 0.28, p = 0.436$ and $3.64 \pm 0.32 \text{ vs} 4.05 \pm 0.28$, p = 0.358, respectively). Per protocol definitions, subjects with ARFID had lower BMI and BMI percentiles than HC. Total fat mass, as measured by DXA, was lower in ARFID compared to HC. Mean PYY levels trended higher in ARFID versus HC (Table 1). Medical comorbidities and/or medications potentially

Table 1 Demographic and clinical characteristics in ARFID vs. healthy control groups

Clinical characteristics	ARFID (n = 14)	HC (<i>n</i> =20)	p value
Age (years)	16.41±0.96	16.29±0.78	0.954
Race, <i>n</i> (%)			0.525
Asian	1 (7.14)	1 (5.00)	
Black or African American	0	0	
White	11 (78.58)	15 (75.00)	
More than one race	2 (14.28)	4 (20.00)	
Premenarchal	6 (42.86)	4 (20.00)	0.252
Total exposure to estrogen in past 9 months (months)**	4.79±1.20	6.75 ± 3.64	0.183
Tanner stage (breasts)	3.71 ± 0.30	4.05 ± 0.28	0.436
Tanner stage (pubic hair)	3.64±0.32	4.05 ± 0.28	0.358
Weight (kg)	40.54 ± 1.63	56.60 ± 2.44	< 0.001*
Height (cm)	157.36±2.19	161.72 ± 2.04	0.163
BMI (kg/m²)	16.27 ± 0.27	21.45 ± 0.61	< 0.001*
BMI z-score	$-1.71 \pm 0.25 (n = 11)$	$0.29 \pm 0.17 (n = 16)$	< 0.001*
Total fat mass (kg)	11.14±0.59	16.01 ± 1.42	0.010*
Total lean mass (kg)	28.83±1.18	33.90 ± 2.98	0.211
% Body fat	26.40 ± 0.69	31.67 ± 1.10	< 0.001*
PYY (pg/ml)	$98.18 \pm 13.55 (n = 11)$	$71.40 \pm 5.61 \ (n = 15)$	0.055
Total body BMD (g/cm ²)	0.93±0.03	1.00±0.03	0.105
Total body less head BMD (g/cm ²)	0.81±0.09	0.88 ± 0.03	0.091
Lumbar spine BMD (g/cm ²)	0.84±0.03 (n=13)	$0.87 \pm 0.04 \ (n = 18)$	0.570

 $\mathsf{Mean} \pm \mathsf{SEM}$ for all values unless note otherwise

*Significant *p* value \leq 0.05

**Total exposure to estrogen in the past 9 months refers to the number of months of exposure to estrogen at physiologic levels or months of use of oral contraceptives

ARFID Avoidant/restrictive food intake disorder, BMI Body mass index, HC Healthy controls, PYY peptide YY



Fig. 1 Bone mineral density (BMD) Z-scores for the **a** lumbar spine, **b** total body less head and **c** total body in low weight females with ARFID (black bars) and HC subjects (gray bars). **p* = 0.021; ***p* = 0.055



Fig. 2 Correlation between PYY levels and age-adjusted lumbar spine BMD in subjects with ARFID. Among low weight females with ARFID who also had PYY data (n = 10), PYY levels were negatively correlated with age-adjusted lumbar BMD ($\beta = -0.481$, p = 0.032) on multivariate analysis. *BMD* Bone mineral density; *PYY* peptide YY

detrimental to bone density were not prevalent among subjects with ARFID. Four subjects (28%) had asthma without a history of chronic glucocorticoid use (inhaled or systemic). Regarding medications, two subjects (14%) were taking vitamin D supplements, two (14%) were on calcium supplements (with no known deficiencies) and four (28%) on multivitamins. Other medications included anti-depressants, stimulants for the treatment of attention deficit/hyperactivity disorder, a proton pump inhibitor (Omeprazole), and cyproheptadine to increase appetite.

Bone variables are presented in Table 1 and Fig. 1. Total body BMD Z-scores were significantly lower in ARFID vs. HC (-1.41 ± 0.28 vs -0.50 ± 0.25 respectively, p=0.021) and total body less head BMD Z-scores trended lower in

ARFID versus HC (-1.67 ± 0.40 vs -0.74 ± 0.27 respectively, p=0.055). Lumbar BMD Z-scores were numerically lower in ARFID vs. HC; however, this did not reach statistical significance (-0.95 ± 0.35 vs. -0.67 ± 0.23 respectively, p=0.489).

Among low weight females with ARFID who also had PYY data (n=10), PYY levels were negatively correlated with age-adjusted lumbar BMD (β =-0.481, p=0.032) on multivariate analysis (Fig. 2). We did not find correlations between PYY levels and total body BMD or total body less head BMD (p=0.55 and p=0.72 respectively).

Discussion

This study was the first to compare bone health in lowweight females with ARFID versus HC. We found lower total body BMD Z-scores in those with ARFID, suggesting that bone health may be compromised in this disorder. Low BMD carries a long-term risk of fractures as well as a potentially detrimental effect on quality of life [23]. Ultimately, this line of research may lead to further investigation of bone health in ARFID, potentially leading to establishment of guidelines for screening for bone health in young patients with ARFID. Further, consistent with our hypothesis, we found that higher levels of PYY (a hormone that inhibits osteoblast activity) were associated with lower BMD (when adjusted for age) among subjects with ARFID, though this association was noted for the lumbar spine and not for the total body.

Low BMD is an established concern in young females with AN, with up to 52% of adolescents having a BMD Z -score of < -1 at one or more sites. Further,

an increased risk of fractures is a significant concern among these individuals [5]. In contrast, little is known regarding bone health in individuals with ARFID. An earlier study of 134 young males and females (comparing 118 with AN versus 16 with ARFID) did not show significant differences in BMD between the groups, suggesting that individuals with ARFID may be at similar risk for low BMD compared to individuals with AN [10]. Our study demonstrating lower total BMD Z-scores in low-weight subjects with ARFID compared to HC provides further evidence for the finding that low BMD is a concern in ARFID.

Similar to a previous report from our group [17], in the current study, PYY levels in subjects with ARFID did not significantly differ from HC despite lower weight in ARFID, where one would typically expect suppressed levels of hormones that signal satiety. In fact, PYY levels trended higher in low-weight girls with ARFID than in HC. Furthermore, higher PYY levels were associated with lower age-adjusted lumbar BMD in ARFID, consistent with prior findings in AN [11], and with PYY's known inhibitory effect on bone formation. Interestingly, this association was observed for the lumbar spine, but not for total body BMD. Given that the lumbar spine is mostly trabecular bone, while the total body includes both cortical and trabecular bone, but mostly reflects cortical sites, our findings suggest that higher PYY levels in ARFID might have a deleterious impact preferentially at trabecular sites. This finding is supported by previous studies showing a twofold increase in trabecular bone volume but no significant increase in cortical bone in Y2 receptor knockout mice [24], as well as strong inverse correlations between PYY and BMD, particularly at the spine in women with AN [11]. Further investigation and larger studies are needed to confirm these findings.

Low BMD in AN is considered multifactorial and related to lower BMI, lower lean mass [5, 25], hypogonadism [5, 26], Growth Hormone resistance and low IGF-1 [27], high cortisol levels, high PYY levels [28], and low levels of leptin, oxytocin [29], insulin and amylin [30]. Of note, in this study despite no difference in estrogen exposure and no difference in lean mass across groups, total BMD Z-scores among subjects with ARFID were lower than in HC (p = 0.021). Total body less head BMD Z-scores also trended lower, but this was not statistically significant (p = 0.055). Our group has previously reported that low-weight adolescent females with ARFID have low leptin levels and lower IGF-1 Z-scores than HC [31], which could contribute to bone loss. The current study further indicates a potential role for PYY in low BMD in ARFID. Taken together, these data suggest that low BMD in ARFID, as in AN, may be multifactorial with potentially overlapping [17] and also distinctive contributing factors from AN. Further investigation of the pathophysiology of bone loss in ARFID will be important.

Our study has limitations. Given its cross-sectional design, causality cannot be assumed. Our sample size was small and may have underpowered our capability to identify statistically significant differences across groups, such as for lumbar spine BMD and certain clinical characteristics. In addition, not all subjects had PYY levels assessed and not all had lumbar BMD data. Lastly, serum calcium, phosphorus, alkaline phosphatase and 25-hydroxy vitamin D levels were not available for most of the subjects. Thus, larger studies are needed, potentially with broader inclusion criteria for ARFID (e.g. including males and individuals across the weight spectrum). In concordance with pediatric recommendations, we examined only whole body and lumbar spine BMD. A broader approach would be to include additional sites, such as the total hip and femoral neck, as was recently recommended by the International Society For Clinical Densitometry for older adolescents [32]. Future investigation of additional sites and using more advanced imaging for bone microstructure as well as markers of bone turnover will add to our understanding of bone health and fracture risk in individuals with ARFID.

In summary, bone health in low weight individuals with ARFID is of concern. Higher levels of PYY may promote multifactorial bone loss in low weight females with ARFID. Further investigation is warranted to improve our understanding of bone health in individuals with ARFID and to guide monitoring and treatment.

Abbreviations

AN	Anorexia nervosa
ARFID	Avoidant/restrictive food intake disorder
BMI	Body mass index
BMD	Bone mineral density
CV	Coefficient of variation
DXA	Dual X-ray absorptiometry
EDA-5	Eating Disorder Assessment for DSM-5
EDE	Eating Disorder Examination
EBW	Expected body weight
HC	Healthy controls
KSADS-P	L Kiddie Schedule for Affective Disorders and Schizophrenia-present and lifetime
MGH	Massachusetts General Hospital
MA	Massachusetts
NIH	National Institutes of Health
PYY	Peptide YY
PARDI	Pica, ARFID, and Rumination Disorder Interview
SEM	Standard error of the mean

Acknowledgements

Not applicable.

Author contributions

All authors listed have made an intellectual contribution to the work and approved it for publication. ACS collected the data with the help of CS, EA. KH, MK, MS helped with coordination of the study. EAL, MM, JJT directed the implementation of the study. ACS analyzed the data. ACS, EAL, MM and JJT

interpreted the results. ACS drafted the manuscript. EAL, MM, JJT secured funding for the project. CS, EA, EAL, JJT, KH, KRB, KTE, MM, MK, MS, NM read, revised, and approved the manuscript prior to submission.

Funding

R01 MH108595 (JJT, EAL, NM), R01 MH103402 (MM, EAL, KTE), K24 MH120568 (EAL), T32 DK007028 (ACS), K23MH125143 (KRB), 1 UL1 TR002541-01, P30DK040561.

Availability of data and materials

The datasets used and/or analyzed during the current study are available from the corresponding author on reasonable request.

Declarations

Ethics approval and consent to participate

For participants < 18 years old, written consent was signed by a parent/ guardian and assent by the participant. Participants ≥ 18 years signed written consent. All study procedures were approved by the Mass General Brigham Institutional Review Board.

Consent for publication

Not applicable.

Competing interests

JJT and KTE receive royalties from Cambridge University Press for the sale of their book *Cognitive-Behavioral Therapy for Avoidant/Restrictive Food Intake Disorder: Children, Adolescents, and Adults.* JJT, KRB, and KTE receive royalties from Cambridge University Press for their book *The Picky Eater's Recovery Book: Overcoming Avoidant/Restrictive Food Intake Disorder.* EAL has served on the scientific advisory board and has/had a financial interest in OXT Therapeutics, a company that developed oxytocin-based therapeutics to treat obesity and metabolic disease. In addition, EAL received funding for an investigator-initiated study from Tonix Pharmaceuticals. MM has served as a consultant for Abbvie and Sanofi and on the scientific advisory board of Abbvie and Ipsen. Their interests were reviewed and are managed by Massachusetts General Brigham Hospital in accordance with their conflict-of-interest policies. All other co-authors have no conflicts of interest.

Author details

¹Neuroendocrine Unit, Massachusetts General Hospital, 55 Fruit St., Boston, MA 02114, USA. ²Division of Pediatric Endocrinology, Mass General Hospital for Children, Boston, MA, USA. ³Eating Disorders Clinical and Research Program, Massachusetts General Hospital, Boston, MA, USA. ⁴Department of Psychiatry, Harvard Medical School, Boston, MA, USA. ⁵Great Ormond Street Institute of Child Health, University College London, London, UK. ⁶Department of Pediatrics Gynecology and Obstetrics, Faculty of Medicine, University of Geneva, Geneva, Switzerland. ⁷Department of Pediatrics, Harvard Medical School, Boston, MA, USA. ⁸Department of Medicine, Harvard Medical School, Boston, MA, USA. ⁹Mental Health Services in the Capital Region of Denmark, Eating Disorders Research Unit, Psychiatric Centre Ballerup, Ballerup, Denmark.

Received: 31 January 2023 Accepted: 11 June 2023 Published online: 01 July 2023

References

- Nicely TA, Lane-Loney S, Masciulli E, Hollenbeak CS, Ornstein RM. Prevalence and characteristics of avoidant/restrictive food intake disorder in a cohort of young patients in day treatment for eating disorders. J Eat Disord. 2014;2:21. https://doi.org/10.1186/s40337-014-0021-3.
- Fisher MM, Rosen DS, Ornstein RM, Mammel KA, Katzman DK, Rome ES, et al. Characteristics of avoidant/restrictive food intake disorder in children and adolescents: a "new disorder" in DSM-5. J Adolesc Health Off Publ Soc Adolesc Med. 2014;55:49–52. https://doi.org/10.1016/j.jadoh ealth.2013.11.013.
- Thomas JJ, Lawson EA, Micali N, Misra M, Deckersbach T, Eddy KT. Avoidant/restrictive food intake disorder: a three-dimensional model of

neurobiology with implications for etiology and treatment. Curr Psychiatry Rep. 2017;19:54. https://doi.org/10.1007/s11920-017-0795-5.

- Cooney M, Lieberman M, Guimond T, Katzman DK. Clinical and psychological features of children and adolescents diagnosed with avoidant/ restrictive food intake disorder in a pediatric tertiary care eating disorder program: a descriptive study. J Eat Disord. 2018. https://doi.org/10.1186/ s40337-018-0193-3.
- Misra M, Aggarwal A, Miller KK, Almazan C, Worley M, Soyka LA, et al. Effects of anorexia nervosa on clinical, hematologic, biochemical, and bone density parameters in community-dwelling adolescent girls. Pediatrics. 2004;114:1574–83. https://doi.org/10.1542/peds.2004-0540.
- Soyka LA, Misra M, Frenchman A, Miller KK, Grinspoon S, Schoenfeld DA, et al. Abnormal bone mineral accrual in adolescent girls with anorexia nervosa. J Clin Endocrinol Metab. 2002;87:4177–85. https://doi.org/10. 1210/jc.2001-011889.
- Becker KR, Keshishian AC, Liebman RE, Coniglio KA, Wang SB, Franko DL, et al. Impact of expanded diagnostic criteria for avoidant/restrictive food intake disorder on clinical comparisons with anorexia nervosa. Int J Eat Disord. 2019;52:230–8. https://doi.org/10.1002/eat.22988.
- Bachrach LK. Acquisition of optimal bone mass in childhood and adolescence. Trends Endocrinol Metab TEM. 2001;12:22–8. https://doi.org/10. 1016/s1043-2760(00)00336-2.
- Schorr M, Drabkin A, Rothman MS, Meenaghan E, Lashen GT, Mascolo M, et al. Bone mineral density and estimated hip strength in men with anorexia nervosa, atypical anorexia nervosa and avoidant/restrictive food intake disorder. Clin Endocrinol. 2019;90:789–97. https://doi.org/10.1111/cen.13960.
- Alberts Z, Fewtrell M, Nicholls DE, Biassoni L, Easty M, Hudson LD. Bone mineral density in anorexia nervosa versus avoidant restrictive food intake disorder. Bone. 2020;134:115307. https://doi.org/10.1016/j.bone. 2020.115307.
- Utz AL, Lawson EA, Misra M, Mickley D, Gleysteen S, Herzog DB, et al. Peptide YY (PYY) levels and bone mineral density (BMD) in women with anorexia nervosa. Bone. 2008;43:135–9. https://doi.org/10.1016/j.bone.2008.03.007.
- Misra M, Miller KK, Tsai P, Gallagher K, Lin A, Lee N, et al. Elevated peptide YY levels in adolescent girls with anorexia nervosa. J Clin Endocrinol Metab. 2006;91:1027–33. https://doi.org/10.1210/jc.2005-1878.
- Baldock PA, Allison SJ, Lundberg P, Lee NJ, Slack K, Lin E-JD, et al. Novel role of Y1 receptors in the coordinated regulation of bone and energy homeostasis. J Biol Chem. 2007;282:19092–102. https://doi.org/10.1074/ jbc.M700644200.
- Wortley KE, Garcia K, Okamoto H, Thabet K, Anderson KD, Shen V, et al. Peptide YY regulates bone turnover in rodents. Gastroenterology. 2007;133:1534–43. https://doi.org/10.1053/j.gastro.2007.08.024.
- Wong IPL, Driessler F, Khor EC, Shi Y-C, Hörmer B, Nguyen AD, et al. Peptide YY regulates bone remodeling in mice: a link between gut and skeletal biology. PloS One. 2012;7:e40038. https://doi.org/10.1371/journal. pone.0040038.
- Yahara M, Tei K, Tamura M. Inhibition of neuropeptide Y Y1 receptor induces osteoblast differentiation in MC3T3-E1 cells. Mol Med Rep. 2017;16:2779–84. https://doi.org/10.3892/mmr.2017.6866.
- Becker KR, Mancuso C, Dreier MJ, Asanza E, Breithaupt L, Slattery M, et al. Ghrelin and PYY in low-weight females with avoidant/restrictive food intake disorder compared to anorexia nervosa and healthy controls. Psychoneuroendocrinology. 2021;129:105243. https://doi.org/10.1016/j. psyneuen.2021.105243.
- Kaufman J, Birmaher B, Brent D, Rao U, Flynn C, Moreci P, et al. Schedule for affective disorders and schizophrenia for school-age children-present and lifetime version (K-SADS-PL): initial reliability and validity data. J Am Acad Child Adolesc Psychiatry. 1997;36:980–8. https://doi.org/10.1097/ 00004583-199707000-00021.
- Sysko R, Glasofer DR, Hildebrandt T, Klimek P, Mitchell JE, Berg KC, et al. The eating disorder assessment for DSM-5 (EDA-5): development and validation of a structured interview for feeding and eating disorders. Int J Eat Disord. 2015;48:452–63. https://doi.org/10.1002/eat.22388.
- Bryant-Waugh R, Micali N, Cooke L, Lawson EA, Eddy KT, Thomas JJ. Development of the Pica, ARFID, and rumination disorder interview, a multi-informant, semi-structured interview of feeding disorders across the lifespan: a pilot study for ages 10–22. Int J Eat Disord. 2019;52:378–87. https://doi.org/10.1002/eat.22958.

- Fairburn CG, Cooper Z, Oonnor M. Eating disorder examination. In: Fairburn CG, editor. Cognitive behavior therapy and eating disorders. 17th ed. New York: Guilford Press; 2008.
- Zemel BS, Kalkwarf HJ, Gilsanz V, Lappe JM, Oberfield S, Shepherd JA, et al. Revised reference curves for bone mineral content and areal bone mineral density according to age and sex for black and non-black children: results of the bone mineral density in childhood study. J Clin Endocrinol Metab. 2011;96:3160–9. https://doi.org/10.1210/jc.2011-1111.
- Johnston CC, Slemenda CW. Peak bone mass, bone loss and risk of fracture. Osteoporos Int J Establ Result Coop Eur Found Osteoporos Natl Osteoporos Found USA. 1994;4(Suppl 1):43–5. https://doi.org/10.1007/ BF01623435.
- Baldock PA, Sainsbury A, Couzens M, Enriquez RF, Thomas GP, Gardiner EM, et al. Hypothalamic Y2 receptors regulate bone formation. J Clin Invest. 2002;109:915–21. https://doi.org/10.1172/JCl14588.
- Miller KK, Lee EE, Lawson EA, Misra M, Minihan J, Grinspoon SK, et al. Determinants of skeletal loss and recovery in anorexia nervosa. J Clin Endocrinol Metab. 2006;91:2931–7. https://doi.org/10.1210/jc.2005-2818.
- Grinspoon S, Thomas E, Pitts S, Gross E, Mickley D, Miller K, et al. Prevalence and predictive factors for regional osteopenia in women with anorexia nervosa. Ann Intern Med. 2000;133:790–4. https://doi.org/10.7326/0003-4819-133-10-200011210-00011.
- Misra M, Miller KK, Bjornson J, Hackman A, Aggarwal A, Chung J, et al. Alterations in growth hormone secretory dynamics in adolescent girls with anorexia nervosa and effects on bone metabolism. J Clin Endocrinol Metab. 2003;88:5615–23. https://doi.org/10.1210/jc.2003-030532.
- Lawson EA, Donoho D, Miller KK, Misra M, Meenaghan E, Lydecker J, et al. Hypercortisolemia is associated with severity of bone loss and depression in hypothalamic amenorrhea and anorexia nervosa. J Clin Endocrinol Metab. 2009;94:4710–6. https://doi.org/10.1210/jc.2009-1046.
- Lawson EA, Donoho DA, Blum JI, Meenaghan EM, Misra M, Herzog DB, et al. Decreased nocturnal oxytocin levels in anorexia nervosa are associated with low bone mineral density and fat mass. J Clin Psychiatry. 2011;72:1546–51. https://doi.org/10.4088/JCP.10m06617.
- Misra M, Miller KK, Cord J, Prabhakaran R, Herzog DB, Goldstein M, et al. Relationships between serum adipokines, insulin levels, and bone density in girls with anorexia nervosa. J Clin Endocrinol Metab. 2007;92:2046–52. https://doi.org/10.1210/jc.2006-2855.
- Aulinas A, Marengi DA, Galbiati F, Asanza E, Slattery M, Mancuso CJ, et al. Medical comorbidities and endocrine dysfunction in low-weight females with avoidant/restrictive food intake disorder compared to anorexia nervosa and healthy controls. Int J Eat Disord. 2020;53:631–6. https://doi. org/10.1002/eat.23261.
- 32. Weber DR, Boyce A, Gordon C, Högler W, Kecskemethy HH, Misra M, et al. The utility of DXA Assessment at the forearm, proximal femur, and lateral distal femur, and vertebral fracture assessment in the pediatric population: 2019 ISCD official position. J Clin Densitom Off J Int Soc Clin Densitom. 2019;22:567–89. https://doi.org/10.1016/j.jocd.2019.07.002.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

